Author's response to reviews

Title: Effects of a grape-supplemented diet on proliferation and Wnt signaling in the colonic mucosa are greatest for those over age 50 and with high arginine consumption

Authors:

Randall F Holcombe (randall.holcombe@mssm.edu)
Micaela Martinez (micealam@uci.edu)
Kestutis Planutis (kiastoutis.planoutis@mssm.edu)
Marina Planutiene (marina.planoutene@mssm.edu)

Version: 2 Date: 28 May 2015

Author's response to reviews:

Please see revision letter for detailed responses to all of the reviewer's comments.